Author’s response to reviews

Title: The association between daily concentrations of air pollution and visits to a psychiatric emergency unit: A case-crossover study

Authors:

Anna Oudin (anna.oudin@med.lu.se)
Daniel Åström (daniel.oudin.astrom@med.lu.se)
Peter Asplund (peter.asplund@vgregion.se)
Steinn Steingrimsson (steinn.steingrimsson@vgregion.se)
Zoltan Szabo (zoltan.szabo@vgregion.se)
Hanne Carlsen (hannekcarlsen@gmail.com)

Version: 2 Date: 07 Dec 2017

Author’s response to reviews:

Dear Editor and Reviewer 1,

Thank you for considering our revised version of the manuscript.

We have rephrased the sentence in the Discussion referring to Zeger et al., and done a few cosmetic changes to the manuscript.

Especially, we want to take the opportunity to thank Reviewer 1 for his very insightful comments regarding measurement error, which substantially improved our manuscript.

Yours sincerely,

Anna Oudin on behalf of all co-authors